Author's response to reviews

Title: Zinc sulfate as an adjunct to methylphenidate for the treatment of attention deficit hyperactivity disorder in children: A double blind and randomized trial

Authors:

Dr Shahin Akhondzadeh (s.akhond@neda.net)  
Mohammad-Reza Mohammadi (scientwr@sina.tums.ac.ir)  
Mojghan Khademi (a3057@sina.tums.ac.ir)

Version: 2 Date: 14 Feb 2004

February 16, 2004

BMC Psychiatry

Dear Iratxe Puebla

I would like to submit our revised version of manuscript (1385495598290715) entitled: Zinc sulfate as an adjunct to methylphenidate for the treatment of attention deficit hyperactivity disorder in children: A double blind and randomized trial. We considered all queries and comments of reviewers point to point as follow:

Reviewer 1 (Dr. Steven Pliszka):
Minor Essential Revisions
1- Labels for figures were written
2- Queries 1-4: The requests of reviewer were considered in the revision

Reviewer 2 (Dr. William C Heird)
Major Compulsory Revision

Query 1: We reduced the statistical section as reviewer comments
Query 2: The reviewer's request was considered in the revision
Query 3: The reviewer's request was considered in the revision
Query 4: The reviewer's request was considered in the revision
Query 5: The reviewer's request was considered in the revision
Query 6: The reviewer's request was considered in the revision

Minor essential Revisions:

Query 1-9: All reviewer's requests were considered in the revision

Discretionary Revisions:
Comments 1, 2, 5, 6, 7, 8 and 9 were considered in the revision.

Sincerely yours

Shahin Akhondzadeh Ph.D.
Associate Professor, Clinical Neuropsychopharmacology
Roozbeh Psychiatric Hospital